\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage[authoryear]{natbib}



\newif\ifnokeywords\nokeywordsfalse

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}
\let\@issuelogo\@empty%
\let\@issuelogoex\@empty%
\def\issuelogo#1{\gdef\@issuelogo{#1}}
\def\issuelogoex#1{\gdef\@issuelogoex{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}
\lefthyphenmin = 3

\def\title#1{%
  \gdef\@title{%
      \vspace*{-40pt}%
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%%{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}\hfill
         \mbox{~}\hspace{5pt}\parbox{.1\linewidth}{\IfFileExists{images/\@issuelogo.\@issuelogoex}{\includegraphics[width=79pt,height=61pt]{images/\@issuelogo.\@issuelogoex}}{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}}\hfill
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
          \begin{center}\fontsize{17}{17}\selectfont\scshape\vskip-7pt International Journal of Research in Pharmaceutical Sciences\hfill\end{center}%
          \vspace*{-10pt}\hspace*{4pt}{\fontsize{8}{9}\selectfont Published by JK Welfare \& Pharmascope Foundation\hfill  Journal Home Page: \href{https://ijrps.com}{\color{blue}\underline{\smash{https://ijrps.com}}}}\hspace*{4pt}\mbox{}}}%
       \par\vspace*{-1pt}\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}

\def\abbrvJournalTitle{Int. J. Res. Pharm. Sci.}

\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}

                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
%\def\bibjtitle#1{#1}   %%Publisher request
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}

\makeatother
\date{}


\usepackage{float}

\begin{document}


\def\RunningAuthor{Srikanth Kanithi et al.}
\firstPage{3674}
\articleType{Original Article}
\receivedDate{03.07.2019}
\acceptedDate{27.10.2019}
\revisedDate{17.10.2019}
\journalVolume{10}
\journalIssue{4}
\journalDoi{ijrps.v10i4.1753}
\copyrightYear{2019}
\def\authorCount{3}
\def\affCount{1}

\def\journalTitle{International Journal of Research in Pharmaceutical Sciences}

\title{Emtricitabine, Tenofovir, and Rilpivirine from their degradation products Analysis by HPLC}
\author{Srikanth~Kanithi,  
            Chebolu~Naga~Sesha~Sai~Pavan~Kumar\textsuperscript{*}, 
            Thulaseedhar~Alumuri~\\[5pt]{Department of Sciences and Humanities, Chemistry Division, Vignan's Foundation for Science, Technology \& Research, Vadlamudi, Guntur-522 213, Andhra Pradesh, India}}

\begin{abstract}
A simple, expeditious, and explicit stability-indicating High-Performance Liquid Chromatography (HPLC) analytical test method was developed for the quantitative analysis of Emtricitabine, Tenofovir, and Rilpivirine in bulk drugs and combined dosage formulations. Using the ICH guidelines method was validated using a C18 column of size 250 mm X 4.6 mm, 5\ensuremath{\mu}, maintained at 25\ensuremath{^{o}}C. The total run time maintained was 10 min. Acetonitrile and 0.1\% TEA prepared in water adjusted with pH 3.0 was adapted as a mobile phase. The injection volume of samples was 20\ensuremath{\mu }L and UV-DAD detector system set at 265 nm used for UV detection. As per the ICH guidelines method was validated. The retention times were observed as 2.52, 3.27, 6.70 min for Emtricitabine, Rilpivirine, and Tenofovir disoproxyl fumarate, respectively. Linearity ranges were observed 24-56 \ensuremath{\mu}g/mL Emtricitabine, 3-7 \ensuremath{\mu}g/mL Rilpivirine and 30-70 \ensuremath{\mu}g/mL Tenofovir. Relative Standard Deviation not exceed 2\%. In contract to the conventional method of HPLC, a method developed was able to give better resolution of swift retention times in the separation of various degradation products along with the pure active pharmaceutical ingredients. The proposed method exhibited excellent reproducibility and repeatability. The stress studies performed by following ICH guidelines indicated that the present HPLC method is explicit and stability-indicating. Since the present HPLC method has the capacity to separate the drugs with high resolution in tablet dosage forms, hence the method can be exploited for routine analysis of quality control sample and stability analysis.
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}RP-HPLC,\newline Stability Indicating Assay,\newline Emtricitabine,\newline Tenofovir,\newline Rilpivirine Forced Degradation Studies
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright\textsuperscript{*}Corresponding Author\par Name:\ Chebolu~Naga~Sesha~Sai~Pavan~Kumar~\\ Phone:\ +91 8500004626~\\  Email:\ pavaniict@gmail.com
        \par}\vspace*{-11pt}\hrulefill\par{\fontsize{12}{14}\selectfont ISSN: 0975-7538}\par% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-11pt}\hrulefill\\{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\par{IJRPS |~\href{https://ijrps.com}{https://ijrps.com}}\par%
        \vspace*{-2pt}{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\par%
        \vspace*{-11pt}\rule{\linewidth}{1.2pt}
        \makeatother
        
    
\section{Introduction}
\textbf{Emtricitabine}

Emtricitabine (EMCB) with chemical name 6-(3-Chloro-2-fluorobenzyl)-1-[(2S)-1 hydroxy-3-methylbutan-2-yl]-7-methoxy-4-oxo-1, 4-dihydroquinoline-3-carboxylic acid. Emtricitabine is the (-) - enantiomer of a thioanalog of cytidine, which differs from other cytidine analogs in that has a fluorine in the 5-position as shown in Figure~\ref{f-2a877c749666}. 

In adults, the Emtricitabine is used for the HIV infection treatment and is an inhibitor of a nucleoside reverse transcriptase inhibitor (NRTI) \unskip~\citep{625702:14635374}. C\ensuremath{_{8}}H\ensuremath{_{10}}FN\ensuremath{_{3}}O\ensuremath{_{3}}S, and 247.25 are the molecular formula and molecular weight of Emtricitabine respectively. The approximate solubility of Emtricitabine in the water at 25 \ensuremath{^\circ}C is 112mg/ml, and physical appearance is a crystalline powder with white to off-white colour.

\textbf{Rilpivirine}

Rilpivirine with chemical name 4-[[4-[4-[(\textit{E})-2-cyanoethenyl]-2,6-dimethylanilino] Pyrimidin-2-yl] amino] benzonitrile shown in Figure~\ref{f-921367932164}. Rilpivirine (TMC278, Trade name Edurant) \unskip~\citep{625702:14635375} used for the treatment of HIV infection and is a second-generation non-nucleoside reverse transcriptase inhibitor (NNRTI) of with higher potency, longer half-life and reduced side-effect profile compared with older NNRTIs, such as Efavirenz. The molecular weight and its molecular formula of Rilpivirine are 366.43 and C\ensuremath{_{22}} H\ensuremath{_{18}} N\ensuremath{_{6,\ }} respectively. The appearance of Rilpivirine is a white to off-white crystalline powder, and the solubility Rilpivirine is 1.62 mg/mL in the water at 25\ensuremath{^\circ}C.


\bgroup
\fixFloatSize{images/d7832547-eb36-4ab6-bc48-87cf8d92d62f-upicture1.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/d7832547-eb36-4ab6-bc48-87cf8d92d62f-upicture1.png}{\includegraphics{images/d7832547-eb36-4ab6-bc48-87cf8d92d62f-upicture1.png}}{}
\makeatother 
\caption{\boldmath {Chemical Structure of Emtricitabine}}
\label{f-2a877c749666}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/5f9246e2-e072-4071-917e-cca3ce514bf7-upicture2.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/5f9246e2-e072-4071-917e-cca3ce514bf7-upicture2.png}{\includegraphics{images/5f9246e2-e072-4071-917e-cca3ce514bf7-upicture2.png}}{}
\makeatother 
\caption{\boldmath {Chemical Structure of Rilpivirine}}
\label{f-921367932164}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/07d7ee6a-63fa-4f14-a358-80a8c4de2224-upicture3.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/07d7ee6a-63fa-4f14-a358-80a8c4de2224-upicture3.png}{\includegraphics{images/07d7ee6a-63fa-4f14-a358-80a8c4de2224-upicture3.png}}{}
\makeatother 
\caption{\boldmath {Chemical Structure of Tenofovirdisoproxilfumarate}}
\label{f-632c5ab16f44}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/cb84f1e3-ff5b-4781-9d45-3248bb3ce0ae-upicture4.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/cb84f1e3-ff5b-4781-9d45-3248bb3ce0ae-upicture4.png}{\includegraphics{images/cb84f1e3-ff5b-4781-9d45-3248bb3ce0ae-upicture4.png}}{}
\makeatother 
\caption{\boldmath {Calibration curve for Emtricitabine}}
\label{f-7fa14caaa9c7}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/8a6c2b17-57f1-4d4f-b951-ef961469960d-upicture5.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/8a6c2b17-57f1-4d4f-b951-ef961469960d-upicture5.png}{\includegraphics{images/8a6c2b17-57f1-4d4f-b951-ef961469960d-upicture5.png}}{}
\makeatother 
\caption{\boldmath {Calibration curve for Rilpivirine}}
\label{f-391ec2ae9e05}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/93c7d22d-eba4-4d16-ad78-61d67bf8a91d-upicture6.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/93c7d22d-eba4-4d16-ad78-61d67bf8a91d-upicture6.png}{\includegraphics{images/93c7d22d-eba4-4d16-ad78-61d67bf8a91d-upicture6.png}}{}
\makeatother 
\caption{\boldmath {Calibration curve for Tenofovir}}
\label{f-20c94d2a77b0}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/3a568ec5-db42-45aa-8ba2-296c88aa7d98-upicture7.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/3a568ec5-db42-45aa-8ba2-296c88aa7d98-upicture7.png}{\includegraphics{images/3a568ec5-db42-45aa-8ba2-296c88aa7d98-upicture7.png}}{}
\makeatother 
\caption{\boldmath {Emtricitabine, Rilpivirine, Tenofovir Chromatogram}}
\label{f-58fd4e683457}
\end{figure}
\egroup
\textbf{Tenofovir disoproxil fumarate}

Tenofovir disoproxil fumarate chemical name of Tenofovir DF is [[(2R)-1-(6-aminopurin-9-yl) propan-2-yl] oxymethyl-(propan-2-yloxycarbonyloxymethoxy) phosphoryl] oxymethylpropan-2-yl carbonate (\textit{E})-but-2-enedioic acid shown in Figure~\ref{f-632c5ab16f44}. Tenofovir disoproxyl fumarate \unskip~\citep{625702:14635376} is known as nucleotide analogue reverse transcriptase inhibitors (NRTIs) which belongs to an anti-retroviral class of drug. Molecular formula and molecular weight of Tenofovir disoproxil fumarate C\ensuremath{_{19}}H\ensuremath{_{30}} N\ensuremath{_{5}}O\ensuremath{_{10}} P\textbf{. }C\ensuremath{_{4}}H\ensuremath{_{4}}O\ensuremath{_{4\ }} and 635.51. The physical appearance of this drug is a white to off-white crystalline powder with a solubility of 13.4 mg/mL in the water at 25 \ensuremath{^\circ}C.
    
\section{Materials and Methods}
\textbf{Drug substance}

Working standards Emtricitabine (99.7\%), Rilpivirine (99.5\%), and Tenofovir disoproxyl fumarate (99.5\%) were procured from Chandra Labs, Hyderabad, India.

\textbf{Instrumentation}

An Agilent-1100, High-Performance Liquid Chromatography consisting of quaternary pump (G1376A), Column compartment (G1315B), Autosampler (G1387A), and Diode array detector (G1315B), supplied by M/s. Agilent Technologies, USA. Mettler-Toledo analytical balance, model AG-245 capable of weighing 0.01 mg, supplier: M/s. Mettler AG, Switzerland. Sonicator supplied by M/s. Serwell instrument, India. Digital pH meter supplied by M/s. Serwell instruments, India.

\textbf{Chemicals and reagents}

Acetonitrile (HPLC grade) was purchased from J.T Bekar, India. Triethyl amine (AR grade) supplied by M/s. Sigma-Aldrich, Bangalore, India. Water (HPLC grade) supplied by M/s. Sigma-Aldrich, India. Sodium hydroxide and Hydrogen peroxide (GR grade) supplied by M/s. Merck, India.

\textbf{Preparation of the mobile phase}

Mobile Phase composition: A{\textendash} Acetonitrile, B{\textendash}0.1\% TEA prepared in water (pH 3.0).

\textbf{Standard solution preparation}

Weigh, and transferred 13 mg of Emtricitabine, 1.62 mg of Rilpivirine, and 20 mg of Tenofovir standards into a 100 mL volumetric flask added 3/4\ensuremath{^{th}} volume of diluent, sonicated for 5 minutes till the sample was dissolved and made up to the final volume with diluent. Prepared 34.62 \ensuremath{\mu}g /mL of solution by diluting 1mL to 10mL with diluent.

\textbf{Chromatographic conditions}

The mixture of A{\textendash} Acetonitrile, and B{\textendash}0.1\% TEA prepared in water (pH 3.0) used as Mobile Phase. The mobile phase was filtered through 0.45\ensuremath{\mu }m membrane filter, degassed for 10 min, and flushed the column with a mobile phase with flow rate, 1.0 mL/min in C18 column, (250mm X 4.6 mm, 5\ensuremath{\mu}). The total run time used was 10min, and the temperature of the column was set at 25\ensuremath{^{o}}C. The volume of the sample injected was 20\ensuremath{\mu }L. Wavelength was set at 265nm in a UV-DAD detector.

\textbf{Method development}

After a number of experiments, prime chromatographic conditions were fixed for better separations \unskip~\citep{625702:14635377,625702:14635378}. For each drug, separate standard calibration lines were erected. From the stock solution, various aliquots were prepared with the concentration ranging from 24-56\ensuremath{\mu}g/mL Emtricitabine, 30-70 \ensuremath{\mu}g/mL Tenofovir, 3-7 \ensuremath{\mu}g/mL Rilpivirine were prepared using the mobile phase. Each concentration was injected 6 times, and for all three drugs, retention time and peak area were recorded each time separately. Construction of Calibration curves for all three drugs were done separately by taking the average peak area on Y-axis and concentration on X-axis. The regression equations, as shown in Figures~\ref{f-7fa14caaa9c7} and~\ref{f-391ec2ae9e05} and Figure~\ref{f-20c94d2a77b0}, were calculated from the calibration curves. The drug content of the combined tablet dosage form was calculated from the above regression equations.

\textbf{Pharmaceutical formulation estimation}

For Solid drug dosage form analysis \unskip~\citep{625702:14635379,625702:14635380}, 10 tablets were weighed, powdered in a mortar and 200mg of powder Emtricitabine, 25 mg of Rilpivirine and 300 mg of Tenofovir taken into a 50 mL volumetric flask. The whole mixture was dissolved in a sufficient quantity of diluent, and finally, the solution was diluted to 50ml by adding the same diluent up to the mark. The solution was subjected to sonication for 5 min. Using 0.45 mm nylon membrane, the solution was filtered. From the above solution, 34.62 \ensuremath{\mu }g/mL of each EMCB, TDF, and Rilpivirine was prepared by dissolving 1 mL of the above solution into 10 mL volumetric flasks and diluted with the mobile phase. Injected 6 times into the column, chromatograms and respective peak areas were measured. The content of EMCB, TDF, and Rilpivirine were calculated by using the regression equation, which was indicated as \% Assay. The results are tabulated in Table~\ref{tw-26f3bab493c7}.


\begin{table*}[!htbp]
\caption{\boldmath {Estimation of Pharmaceutical Formulation} }
\label{tw-26f3bab493c7}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.19169999999999998\linewidth-2\tabcolsep}p{\dimexpr.1875\linewidth-2\tabcolsep}p{\dimexpr.1834\linewidth-2\tabcolsep}p{\dimexpr.2374\linewidth-2\tabcolsep}p{\dimexpr.20\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Drug name & Labelled claim (mg)  & Test Conc.  \mbox{}\protect\newline (\ensuremath{\mu }g/mL)  & Mean Amount estimated(\ensuremath{\mu }g/mL)  & Estimated amount  \mbox{}\protect\newline (\%) \\
\tblmidrule 
Emtricitabine &
  200 &
  200 &
  200.08 &
  100.04\\
Tenofovir &
  300 &
  300 &
  307.23 &
  102.41\\
Rilpivirine &
  25 &
  25 &
  24.93 &
  99.74\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Results of the Recovery Studies } }
\label{tw-3ecbcd56d379}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.15540000000000003\linewidth-2\tabcolsep}p{\dimexpr.15770000000000001\linewidth-2\tabcolsep}p{\dimexpr.1669\linewidth-2\tabcolsep}p{\dimexpr.1808\linewidth-2\tabcolsep}p{\dimexpr.18299999999999997\linewidth-2\tabcolsep}p{\dimexpr.15620000000000001\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Drugs name & Pre analysed Conc.  \mbox{}\protect\newline (\ensuremath{\mu}g/mL)  & \% Recovery levels  & Added Amount (\ensuremath{\mu}g/mL)  & Found Amount   \mbox{}\protect\newline (\ensuremath{\mu}g/mL)(n=3)  & \%  \mbox{}\protect\newline Recovery \\
\tblmidrule 
\multicolumn{1}{p{\dimexpr(.15540000000000003\linewidth-2\tabcolsep)}}{\multirow{3}{\linewidth}{Emtricitabine}} &
  \multicolumn{1}{p{\dimexpr(.15770000000000001\linewidth-2\tabcolsep)}}{\multirow{3}{\linewidth}{40}} &
  80 &
  32 &
  39.7 &
  124.1\\
 &
   &
  100 &
  40 &
  48.0 &
  120.1\\
 &
   &
  120 &
  48 &
  55.6 &
  115.8\\
\multicolumn{1}{p{\dimexpr(.15540000000000003\linewidth-2\tabcolsep)}}{\multirow{3}{\linewidth}{Rilpivirine }} &
  \multicolumn{1}{p{\dimexpr(.15770000000000001\linewidth-2\tabcolsep)}}{\multirow{3}{\linewidth}{05}} &
  80 &
  04 &
  05.0 &
  125.8\\
 &
   &
  100 &
  05 &
  06.1 &
  122.1\\
 &
   &
  120 &
  06 &
  07.0 &
  115.9\\
\multicolumn{1}{p{\dimexpr(.15540000000000003\linewidth-2\tabcolsep)}}{\multirow{3}{\linewidth}{Tenofovir }} &
  \multicolumn{1}{p{\dimexpr(.15770000000000001\linewidth-2\tabcolsep)}}{\multirow{3}{\linewidth}{50}} &
  80 &
  40 &
  49.0 &
  122.6\\
 &
   &
  100 &
  50 &
  60.9 &
  121.7\\
 &
   &
  120 &
  60 &
  69.3 &
  115.5\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Method Precision} }
\label{tw-d7b10cb7a15c}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1103\linewidth-2\tabcolsep}p{\dimexpr.1697\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.16\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}{\multirow{2}{*}{S. No.}} & \multicolumn{2}{p{\dimexpr(.3097\linewidth-2\tabcolsep)}}{\cAlignHack Emtricitabine } & \multicolumn{2}{p{\dimexpr(.28\linewidth-2\tabcolsep)}}{\cAlignHack Rilpivirine } & \multicolumn{2}{p{\dimexpr(.30\linewidth-2\tabcolsep)}}{\cAlignHack Tenofovir}\\
\rowcolor{kwdboxcolor} & RT  & Area  & RT  & Area  & RT & Area\\
\tblmidrule 
1 &
  2.520 &
  1087.803 &
  3.277 &
  1094.372 &
  6.707 &
  1753.454\\
2 &
  2.517 &
  1089.666 &
  3.270 &
  1101.764 &
  6.690 &
  1774.741\\
3 &
  2.523 &
  1067.836 &
  3.277 &
  1073.595 &
  6.707 &
  1729.087\\
4 &
  2.520 &
  1097.279 &
  3.273 &
  1108.428 &
  6.707 &
  1774.492\\
5 &
  2.517 &
  1059.014 &
  3.270 &
  1079.236 &
  6.693 &
  1729.167\\
6 &
  2.517 &
  1071.854 &
  3.273 &
  1075.584 &
  6.693 &
  1727.320\\
Avg &
  2.5190 &
  1078.909 &
  3.273 &
  1088.830 &
  6.700 &
  1748.044\\
SD &
  0.0024 &
  14.836 &
  0.003 &
  14.708 &
  0.008 &
  22.745\\
\%RSD &
  0.10 &
  1.38 &
  0.10 &
  1.35 &
  0.12 &
  1.30\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Calibration Data} }
\label{tw-1cd128595e79}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.15080000000000002\linewidth-2\tabcolsep}p{\dimexpr.1692\linewidth-2\tabcolsep}p{\dimexpr.16\linewidth-2\tabcolsep}p{\dimexpr.16\linewidth-2\tabcolsep}p{\dimexpr.16\linewidth-2\tabcolsep}p{\dimexpr.20\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}\multicolumn{2}{p{\dimexpr(.32\linewidth-2\tabcolsep)}}{\cAlignHack Emtricitabine} & \multicolumn{2}{p{\dimexpr(.32\linewidth-2\tabcolsep)}}{\cAlignHack Rilpivirine } & \multicolumn{2}{p{\dimexpr(.36\linewidth-2\tabcolsep)}}{\cAlignHack Tenofovir}\\
\rowcolor{kwdboxcolor}Mcg  & Area  & Mcg  & Area  & Mcg & Area\\
\tblmidrule 
24 &
  631.586 &
  3 &
  659.236 &
  30 &
  1229.584\\
32 &
  907.713 &
  4 &
  919.393 &
  40 &
  1482.509\\
40 &
  1091.004 &
  5 &
  1086.050 &
  50 &
  1750.266\\
48 &
  1339.312 &
  6 &
  1348.518 &
  60 &
  2124.626\\
56 &
  1549.123 &
  7 &
  1552.332 &
  70 &
  2413.579\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Optical Characteristics of Emtricitabine, Rilipivirine and Tenofovir} }
\label{tw-13efc19c1c96}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.2725\linewidth-2\tabcolsep}p{\dimexpr.22790000000000003\linewidth-2\tabcolsep}p{\dimexpr.26039999999999996\linewidth-2\tabcolsep}p{\dimexpr.2392\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Parameters & Emtricitabine  & Rilpivirine  & Tenofovir \\
\tblmidrule 
Linearity range (\ensuremath{\mu}g/mL) &
  24-56 &
  3-7 &
  30-70\\
Regression line equation &
  y = 28.33x+29.589 &
  y = 221.53x+5.4473 &
  y = 30.101x+295.06\\
Correlation coefficient (r) &
  0.9969 &
  0.9963 &
  0.9951\\
LOD (\ensuremath{\mu}g/mL) &
  1.47 &
  0.28 &
  4.16\\
LOQ (\ensuremath{\mu}g/mL) &
  4.46 &
  0.86 &
  12.61\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {LOD \& LOQ} }
\label{tw-4f9c6d55d59b}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.12219999999999999\linewidth-2\tabcolsep}p{\dimexpr.138\linewidth-2\tabcolsep}p{\dimexpr.1598\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.16\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor} & \multicolumn{2}{p{\dimexpr(.2978\linewidth-2\tabcolsep)}}{\cAlignHack Emtricitabine } & \multicolumn{2}{p{\dimexpr(.28\linewidth-2\tabcolsep)}}{\cAlignHack Rilpivirine } & \multicolumn{2}{p{\dimexpr(.30\linewidth-2\tabcolsep)}}{\cAlignHack Tenofovir }\\
\rowcolor{kwdboxcolor} & Mcg  & Area  & Mcg  & Area  & Mcg  & Area \\
\tblmidrule 
LOD &
  1.47 &
  41.81 &
  0.28 &
  62.73 &
  4.16 &
  125.55\\
LOQ &
  4.46 &
  126.70 &
  0.86 &
  190.10 &
  12.61 &
  380.47\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Robustness Studies of Emtricitabine, Rilpivirine, and Tenofovir} }
\label{tw-925aa8cb924e}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.2083\linewidth-2\tabcolsep}p{\dimexpr.1903\linewidth-2\tabcolsep}p{\dimexpr.2222\linewidth-2\tabcolsep}p{\dimexpr.2028\linewidth-2\tabcolsep}p{\dimexpr.1764\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}{\multirow{2}{*}{Method parameters}} & {\multirow{2}{*}{Conditions }} & \multicolumn{3}{p{\dimexpr(.6014\linewidth-2\tabcolsep)}}{\cAlignHack Retention Time (RT) }\\
\rowcolor{kwdboxcolor} &  & Emtricitabine  & Rilpivirine  & Tenofovir \\
\tblmidrule 
Flow Rate &
  1.4mL/min &
  2.99 &
  3.88 &
  7.89\\
Flow Rate &
  1.0mL/min &
  2.17 &
  2.81 &
  5.70\\
Wavelength &
  260nm &
  2.49 &
  3.24 &
  6.63\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {System Suitability Parameters} }
\label{tw-3a6662e21c6b}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.26209999999999997\linewidth-2\tabcolsep}p{\dimexpr.23790000000000003\linewidth-2\tabcolsep}p{\dimexpr.2448\linewidth-2\tabcolsep}p{\dimexpr.2552\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Parameter & Emtricitabine  & Rilpivirine  & Tenofovir \\
\tblmidrule 
Retention time &
  2.99 &
  3.88 &
  7.89\\
Asymmetry &
  1.77 &
  1.31 &
  1.609\\
Plates &
  3559 &
  2418 &
  5817\\
Resolution &
  - &
  2.90 &
  11.09\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Results of stress degradation studies} }
\label{tw-13da8e42a494}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.26549999999999997\linewidth-2\tabcolsep}p{\dimexpr.2968\linewidth-2\tabcolsep}p{\dimexpr.2223\linewidth-2\tabcolsep}p{\dimexpr.2154\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}{\multirow{2}{*}{Stress conditions}} & \multicolumn{3}{p{\dimexpr(.7344999999999999\linewidth-2\tabcolsep)}}{\cAlignHack Active present after degradation (\%) }\\
\rowcolor{kwdboxcolor} & Emtricitabine & Rilpivirine & Tenofovir\\
\tblmidrule 
Acid &
  27.7 &
  28.0 &
  44.3\\
Base &
  27.5 &
  27.6 &
  44.9\\
Thermal &
  27.6 &
  27.7 &
  44.7\\
Photolytic &
  27.8 &
  27.9 &
  44.3\\
Photocatalytic &
  27.7 &
  27.6 &
  44.7\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}
\textbf{Validation}

In this analytical method, validation proved repeatability reproducibility, and accuracy of the method and will steadily harvest a result meeting its intended analytical applications \unskip~\citep{625702:14635381}. ICH guideline was adapted for validation for this method, which provided the information system suitability, linearity, and limit of detection (LOD), the limit of quantification (LOQ), precision, selectivity, robustness, and accuracy. 

\textbf{Accuracy}

The recoveries of each drug at three different levels was done, ranging from (80\% to 120\%) with target concentration EMCB, TDF, and Rilpivirine. The recovery range was tabulated, as indicated in Table~\ref{tw-3ecbcd56d379}. 

\textbf{Precision}

The system precision was tested by six (n=6) injections, and the assay value, \% RSD, RT, and areas were determined in Table~\ref{tw-d7b10cb7a15c}. Assay value calculated, and the obtained \% RSD were 1.38\%, 1.35\%, and 1.30\% respectively. The \%RSD value were found to be {\textless}2.0\%. Statistical results revealed that the proposed HPLC method having worthy precision. The results are shown in tabular form, \textit{i.e.,}Table~\ref{tw-d7b10cb7a15c}.

\textbf{Linearity and Range}

The linearity and range of the method was established in the range of 24-56\ensuremath{\mu }g/mL for Emtricitabine, 30-70\ensuremath{\mu }g/mL for Tenofovir disoproxil fumarate, and 3-7\ensuremath{\mu }g/mL for Rilpivirine concentrations. The graph plotted peak area versus concentration. The data were statistically analyzed by using linear regression. The results were tabulated in Table~\ref{tw-1cd128595e79} and Table~\ref{tw-13efc19c1c96}.

\textbf{LOD and LOQ}

Limit of detection (LOD) and Limit of quantification (LOQ) represents the concentration of the analyte that would yield signal to noise ratio of 3 for LOD and 10 for LOQ, respectively. The results were tabulated in Table~\ref{tw-4f9c6d55d59b}.

\textbf{Robustness}

The assay method's robustness was set up by introducing small changes in the chromatographic condition, which included the percentage of wavelength (260nm and 264nm), 1.0mL/min, and 1.4 mL/min of flow rates. The results were tabulated in Table~\ref{tw-925aa8cb924e}.

\textbf{Solution stability}

The solution stability \unskip~\citep{625702:14635382,625702:14635383} of Emtricitabine, Rilpivirine, and Tenofovir for the assay method was conceded out by parting both the sample and standard solutions in compactly capped volumetric flasks at room temperature for 24 h. The same sample solution was surveyed at 6-hour intervals over the study period. The RSD\% of the Emtricitabine, Rilpivirine, and Tenofovir assay was calculated.

\textbf{System suitability parameters}

For gaging system suitability, six replicates of standards samples of Emtricitabine, Rilpivirine, and Tenofovir were injected and various parameters, \textit{i.e.,} resolution, plate number (N), peak asymmetry and relative retention time of the samples were premeditated. The results were recorded in Table~\ref{tw-3a6662e21c6b}.

\textbf{Specificity and selectivity}

Specificity is the degree to which the procedure applies to check in each analysis by examining blank matrix samples for any interfering peaks and evaluated with regard to interference due to the presence of any other placebos and proved no interfering peaks within retention time ranges shows the respective chromatogram (Figure~\ref{f-58fd4e683457}).

\textbf{Degradation studies}

\textbf{Stock solution preparation}

In 50 mL volumetric flasks, 200 mg of Emtricitabine, 25mg of Rilpivirine, and 300mg of Tenofovir were weighed. The mixture was dissolved in the mobile phase. The solution was finally diluted to the mark in 50ml volumetric flasks using the same diluent.

\textbf{Acid degradation}

In a 50ml round bottom flask, 5mL of stock solution was pipetted, and 2.5 mL of 0.1 N HCl was added. At 60\ensuremath{^\circ}C, the flask was refluxed for 30 min and then allowed to cool. The solution was neutralized using 0.1N NaOH and made up to the mark using the mobile phase and intended the percentage of degradation.

\textbf{Alkali degradation}

In 50 mL round bottom flask, 5.0 mL of stock solution was pipetted, and 2.5 mL of 0.1 N NaOH was added. The flask was refluxed at 60 \ensuremath{^\circ}C for 30 min and allowed to cool. Neutralized with 0.1N HCl solution, using a mobile phase, the solution was made up to the mark and intended the percentage of degradation.

\textbf{Photocatalytic degradation under UV}

Weighed and transferred 300 mg of Emtricitabine, 50 mg of Rilpivirine, 500 mg of Tenofovir into different Petri dish and kept in UV light. Weighed and transferred 200 mg of Emtricitabine, 25 mg of Rilpivirine, 300 mg of Tenofovir into 50 mL volumetric flasks, dissolved and diluted up to the mark with the mobile phase. In 50 mL volumetric flasks, 5ml stock solution was transferred into and diluted up to the mark with the mobile phase.

\textbf{Thermal condition}

In different Petri dishes, 300 mg of Emtricitabine, 50 mg of Rilpivirine, and 500 mg of Tenofovir were weighed. Dishes were incubated in hot air oven maintained at 105\ensuremath{^\circ}C for 6 h, and the samples were placed in a desiccator till it reaches room temperature. After reaching room temperature again from above plates, 200 mg of Emtricitabine, 25 mg of Rilpivirine, 300 mg of Tenofovir were weighed and transferred into a 50 mL volumetric flask. The mixture was dissolved in the mobile phase and diluted with the same till the mark. In a 50 mL volumetric flasks, 5 mL of each stock solution was transferred into and diluted up to the mark with the mobile phase. 

\textbf{Photolytic condition}

Pipetted, 5 mL stock solution, was bared to sunlight for about 6hours and diluted with 5 mL of the mobile phase. Calculated the percentage of degradation was calculated.
    
\section{Results and Discussion}
\textbf{Chromatographic conditions optimization}

The optimized chromatographic conditions are mentioned above. The finest peak shape and supreme separation were achieved with mobile phase composition is A{\textendash}Acetonitrile, B{\textendash}0.1\% TEA in water with Column: C18 (250 mm x 4.6 mm, 5\ensuremath{\mu}), a wavelength at 265 nm and flow rate of 1.0 mL/min. As per our knowledge till date majority of HPLC methods uses C-18 or C-8 columns and most their mobile phase are complex in compositions. Hence endeavoured were deviated towards the simple method development and enhanced with the C18 column with virtuous resolution. To get good separation, different logical variations were tried among the individual drugs and the solid drug dosages and some modifications in mobile phase composition, flow rate, and temperature in gradient modes on different C18 columns with HPLC. 

\textbf{Accuracy}

The percentage recovery was covered in the range from 80.00\% to 120.00\% for Emtricitabine, Rilpivirine, and Tenofovir in the combined dosage forms. The RSD\% value of replicated injections was less than 2.0\%, which indicates that this method is highly accurate. Table~\ref{tw-3ecbcd56d379} indicates the results obtained.  

\textbf{Precision}

The method precision and intermediate precision of Emtricitabine, Rilpivirine, and Tenofovir standard solution checked, and \%RSD were less than 2\%, which signifies that this method is highly precise. The results obtained are recorded in Table~\ref{tw-d7b10cb7a15c}. 

\textbf{Linearity}

The calibration curve plotted for the Emtricitabine, Rilpivirine, and Tenofovir were linear over the samples concentrations range of 24-56\ensuremath{\mu}g/mL, 3-7\ensuremath{\mu}g/mL, and 30-70\ensuremath{\mu}g/mL, respectively. The data subjected towards the peak area versus concentration by linear regression analysis, and the correlation coefficient (r) was obtained (0.99). The statistical analysis revealed that the proposed method was linear, and Table~\ref{tw-1cd128595e79} \& Table~\ref{tw-13efc19c1c96} indicates the results obtained.

\textbf{LOQ and LOD}

The results of LOQ and LOD data was established for all analytes, and the signal-to-noise ratio for the LOQ and LOD were well within the acceptance criteria. The results were as tabulated in Table~\ref{tw-4f9c6d55d59b}.

\textbf{Robustness}

The robustness of the assay method was established by introducing minor changes in the chromatographic condition, which covered the percentage of flow rate, 1.0 and 1.4 mL/min and wavelength, 260 nm, and 264 nm. The developed method was unaffected by the minor deliberated changes, which indicates the proposed method was robust. The results were tabulated in Table~\ref{tw-925aa8cb924e}.

\textbf{Degradation studies}

Upon performance of degradation studies, Emtricitabine, Rilpivirine, and Tenofovir were active after degradation. The active present after degradation studies results were shown in Table~\ref{tw-13da8e42a494}. 
    
\section{Conclusion}
Stress testing (or forced degradation studies) one of the most critical criteria in pharmaceutical industries in method development. A simple, expeditious, rapid, and precise stability-indicating analytical method of HPLC was developed, and the methods were validated for the quantitative analysis of Emtricitabine, Rilpivirine, and Tenofovir in bulk drugs and combined dosage forms. The better separation of various types of degradation products along with pure drugs can be achieved. The method is also has capacity of giving swift retention times with upright resolution. Excellent performance with respect to Sensitivity, repeatability, and swift can be achieved by this method. Thus method could be useful not only for assay but also for monitoring and degradation products during development as per the ICH guidelines. Hence using this method, routine QC samples and control samples can be analyzed. When compared with other methods, this method will decrease the consumption of solvent, time, man, and power. Hence the method is fit for regular pharmaceutical use. 

\textbf{Acknowledgement}

Authors are very thankful to Vice-chancellor M. Y. S. Prasad, Dean R \& DP. Vidhu, VFSTR for constant support and encouragement. SK thankful to Chandra Labs, Hyderabad, India, for providing Emtricitabine, Rilpivirine, and Tenofovir standards as gift samples and for providing instrumental facilities to conduct the research work.
    

\bibliographystyle{pharmascope_apa-custom}

\bibliography{\jobname}
\end{document}
